Author's response to reviews

Title: The mediating role of resilience in the relationship between stress and life satisfaction among Chinese medical students: a cross-sectional study

Authors:

Meng Shi (815915804@qq.com)
XiaoXi Wang (wxx112@qq.com)
YuGe Bian (gloriabian@icloud.com)
Lie Wang (liewang@mail.cmu.edu.cn)

Version: 3
Date: 6 January 2015

Author's response to reviews: see over
Dear editor


Below is point-by-point description of our revision according to the editor’s comments.

**Editorial Comment:**
There is a disturbing redundancy in reporting of too many decimals in the text and tables that should be dealt:
Please see that all percentages and mean values include only one decimal, all other estimates (betas, correlation coefficients etc) can have two decimals. All P-values can be kept with three decimals, but please substitute P=0.000 with p<0.001.

Response: Relevant changes were made in the whole manuscript as required by the editor. All the changes were marked in red color.

Best regards,

Pro. Lie Wang